Response to reviewers' comments

Editorial comments:

Manuscript:

1. The editor has formatted the manuscript text as per the journal's style. Please retain and use the attached file for revision.

Response:

We retained and used the attached file for revision.

2. Please address all the specific comments marked in the manuscript.

Response:

We addressed all the specific comments marked in the manuscript by directly respond to each comment. All revisions were highlighted by blue font color.

3. The manuscript requires thorough copyediting. Please use professional copyediting services.

Response:

The manuscript was proofread by a professional copyediting service. We attached the invoices for your reference: Proofread invoice 1.pdf and Proofread invoice 2.pdf.

<u>Video:</u>

1. Please make a separate title card for the ethics statement.

Response:

The Ethic statement was separated in a new title card.

2. Please homogenize the narration and the text.

Response:

The narration and the text were homogenized.

3. Please un-italicize "In Vitro" in the title 6:29 -

Response:

The title used italicized "*In Vitro*" according to the previous editorial comments on June 25th, 2021.

4. The title "Conclusion" needs to fade out instead of cutting out. Please fix.

Response:

The conclusion card was fixed by fading out before the next scene.

5. There are three logos that appear at the end of the video. Please remove these as the ASV Criteria mentions not to use any branding.

Response:

All logos which appear at the end of video were removed.